Author's response to reviews

Title: Emerging advantages and drawbacks of Telephone Surveying in Public Health Research in Ireland and the UK

Authors:

Mairin Boland (mairin.boland1@mail.hse.ie)
Mary rose Sweeney (maryrose.sweeney@ucd.ie)
Elaine Scallan (bmd9@cdc.gov)
Mary Harrington (mary.harrington@ucd.ie)
Anthony Staines (anthony.staines@gmail.ie)

Version: 4  Date: 15 August 2006

Author's response to reviews:

UCD School of Public Health and Population Science
UCD
Earlsfort Terrace
Dublin 2.
15/08/06

To: BioMed Central Production Team,

Re: TITLE: Emerging advantages and drawbacks of telephone surveying in public health research in Ireland and the U.K.
M Boland1,2*, MR Sweeney1*, E Scallan1, M Harrington1, A Staines1.

All formatting changes are now complete.

Kind regards

Mary Rose Sweeney